Author’s response to reviews

Title: The introduction of "No jab, No school" policy and the refinement of measles immunisation strategies in high-income countries

Authors:
Filippo Trentini (ftrentini@fbk.eu)
Piero Poletti (poletti@fbk.eu)
Alessia Melegaro (alessia.melegaro@unibocconi.it)
Stefano Merler (merler@fbk.eu)

Version: 4 Date: 15 Mar 2019

Author’s response to reviews:

Dear Editor,

Please find enclosed the revised version of the manuscript “The introduction of "No jab, No school" policy and the refinement of measles immunisation strategies in high-income countries” (BMED-D-18-01492R1).

As suggested, we included for all in-text references to the supplementary material the name of the subsection rather than the page numbers.

Please note that the new version of the manuscript now also adheres to the stylistic guidelines of the journal and we dare to hope that it is now suitable for publication in BMC Medicine.

Sincerely yours,

Filippo Trentini

on behalf of all the authors (Piero Poletti, Alessia Melegaro and Stefano Merler)